Author's response to reviews

Title: Burden of severe RSV disease among immunocompromised children and adults: a 10 year

retrospective study.

**Authors:** 

Olga Chatzis (olga.chatzis@uclouvain.be)

Stephanie Darbre (stephanie.darbre@chuv.ch)

Jérôme Pasquier (jerome.pasquier@chuv.ch)

Pascal Meylan (pascal.meylan@chuv.ch)

Oriol Manuel (oriol.manuel@chuv.ch)

John-David Aubert (john-david.aubert@chuv.ch)

Maja Beck-Popovic (maja.beck-popovic@chuv.ch)

Stavroula Masouridi-Levrat (Stavroula.Masouridi@hcuge.ch)

Marc Ansari (marc.ansari@hcuge.ch)

Laurent Kaiser (laurent.kaiser@hcuge.ch)

Klara Posfay-Barbe (klara.posfay-barbe@hcuge.ch)

Sandra Asner (sandra.asner@chuv.ch)

Version: 4 Date: 12 Feb 2018

**Author's response to reviews:** 

Replies to the reviewers' comments

We thank the editor for his/her comments and for giving us the opportunity to address all minor

edits. Please note that the page and line numbers in our responses to reviewers' comments refer

to our numbering in the revised manuscript.

## **Editor Comments**

Please ensure that all figures/tables and supplementary files are cited within the text. Any items which are not cited may be deleted by our production department upon publication. Table 3 and 4 are not cited within the main text.

We apologize for the inconvenience. Table 3 and 4 have been respectively added p.8, ll.22 and p.9, ll.3)